Author's response to reviews

Title: An ultrasound model to calculate the brain blood outflow through collateral vessels. A pilot study

Authors:

Paolo Zamboni (zambo@unife.it)
Francesco Sisini (ssf@unife.it)
Erica Menegatti (menegatti.eric@qmail.com)
Angelo Taibi (angelo.taibi@unife.it)
Anna M Malagoni (anna.maria.malagoni@unife.it)
Sandra Morovic (sandra_morovic@hotmail.com)
Mauro Gambaccini (mauro.gambaccini@unife.it)

Version: 6 Date: 16 May 2013

Author's response to reviews:

Dear Editor,

as requested we state in the Methods section whether written informed consent for participation in the study was obtained from participants. We added a new paragraph "informed consent" at line 265 to address this issues.

Thank you,

Best Regards,

Paolo Zamboni, MD